RESEARCH ARTICLE



Propensity to consume and the optimality of Ramsey–Euler policies

Tapan Mitra¹ · Santanu Roy²

Received: 16 June 2020 / Accepted: 27 October 2020 / Published online: 9 November 2020 © Springer-Verlag GmbH Germany, part of Springer Nature 2020

Abstract

In a general one-sector optimal stochastic growth model where the production technology may be globally unproductive or allow for unbounded growth, we outline readily verifiable sufficient conditions for optimality that do not require checking the transversality condition. An interior policy function satisfying the Ramsey–Euler condition may not be optimal even if consumption and investment are continuous and increasing in output; our conditions for optimality require that the policy function must also satisfy a lower bound on the propensity to consume. For the case of production functions with multiplicative shocks, the consumption propensity needs to be bounded away from zero; a similar condition is sufficient for more general production functions if the utility function belongs to a restricted class.

Keywords Stochastic growth \cdot Optimal economic growth \cdot Uncertainty \cdot Unbounded growth \cdot Unproductive technology \cdot Transversality condition \cdot Optimality conditions \cdot Euler equation

JEL Classification $C6 \cdot D9 \cdot O41$

T. Mitra: He passed away on February 3, 2019.

Helpful suggestions from an anonymous referee and an Associate Editor of this journal are gratefully acknowledged. This paper has gained from insightful comments received from Debraj Ray, Gerhard Sorger and members of the audience at various presentations including the 2019 Annual Meeting of the SAET in Ischia, Italy.

Santanu Roy sroy@smu.edu

¹ Cornell University, Ithaca, NY, USA

² Department of Economics, Southern Methodist University, 3300 Dyer Street, Dallas, TX 75275-0496, USA

1 Introduction

The one sector model of optimal economic growth under uncertainty (Levhari and Srinivasan 1969; Brock and Mirman 1972) has been widely used by economists to examine problems of capital accumulation in stochastic environments including macroeconomic growth under technology or productivity shocks and resource management under environmental uncertainty. Variations of the model have also been used to study business cycles.

In this model, a representative agent allocates the currently available output (of a single good) between investment and consumption where consumption generates immediate utility while investment generates next period's output according to a production function that is subject to exogenous production shocks. In the standard version of the model, the exogenous shocks are independent and identically distributed over time. The agent maximizes expected discounted sum of utility from consumption where the discount factor, the utility function and the production function are invariant over time. In such a stationary framework, the intertemporal economic trade-offs faced by the agent are reflected in the *optimal* consumption policy function. Conditions for optimality play a very important role in understanding the nature of this optimal policy function. In a large class of applications where economists work with specific functional forms for utility and production functions, sufficient conditions for optimality help determine whether an explicitly specified policy function is actually optimal. Even when one cannot derive explicit solutions to the dynamic optimization problem, sufficient conditions for optimality are useful in showing that a certain implicitly defined ("candidate") function is optimal. Optimality conditions for the dynamic optimization problem underlying the one sector stochastic growth model can also be useful in dynamic games of capital accumulation such as dynamic games of common property renewable resource extraction¹.

In a convex framework (strictly concave utility, concave production function), the existing literature has used duality theory to derive a set of conditions that are both necessary and sufficient for a policy function to be optimal and, in fact, to be the unique optimal policy function. In particular, an interior policy function (i.e., one where both consumption and investment are always strictly positive when the current stock of output is strictly positive) is optimal if, and only if, it satisfies the Euler condition (called the Ramsey–Euler equation in this literature) and a transversality condition (Mirman and Zilcha 1975; Zilcha 1976, 1978).^{2,3}

The Ramsey–Euler equation is a simple first order condition that captures the tradeoff between consumption in any two consecutive time periods, and takes the form of a functional equation. We refer to an interior consumption policy function satisfying

¹ See, for instance, Mitra and Sorger (2014).

² Key contributions emphasizing the importance of the transversality condition in models of intertemporal resource allocation include Malinvaud (1953), Cass (1965), Shell (1969), Peleg and Ryder (1972) and Weitzman (1973).

³ That the Euler and transversality conditions are necessary and sufficient for optimality has been established for more general, convex dynamic optimization problems. See, among others, Stokey and Lucas (1989), Acemoglu (2009). Establishing the necessity of transversality condition for optimality in general has been more challenging; see, Kamihigashi (2001, 2003, 2005).

this Ramsey–Euler equation as a *Ramsey–Euler policy* and this paper contributes to a literature on systematic study of the optimality of such a policy.

Using the characterization results mentioned above, a Ramsey–Euler policy can be shown to be an optimal policy, if it satisfies a transversality condition. The transversality condition essentially requires that the expected present value of capital stocks (valued by a shadow price equal to the discounted marginal utility of current consumption) converges to zero in the long run. It is an asymptotic condition on the entire stochastic process generated by the policy function.⁴ Verifying the transversality condition can be a non-trivial task when the stochastic process of output and consumption can reach levels arbitrarily close to zero infinitely often (for instance, on sample paths involving runs of bad realizations of the production shock) and the marginal utility of consumption is infinitely large at zero⁵; it can also be somewhat challenging if output and investment can be arbitrarily large with positive probability.

Mitra and Roy (2017) develop an alternative sufficient condition for optimality of a Ramsey–Euler policy; they show that a Ramsey–Euler policy function is optimal if it is continuous or co-monotone (i.e., consumption and investment are both non-decreasing in current output). They derive their results under two restrictions on the production technology. First, the technology is assumed to be productive for investment levels close to zero even under the worst realization of the random shock i.e., marginal productivity at zero is always greater than one. Second, the technology is assumed to exhibit bounded growth i.e., there is a maximum sustainable capital stock beyond which the technology is unproductive even for the best realization of the random shock. A natural question that arises is whether their result extends to more general environments where the production technology may be unproductive at all levels of investment or alternatively, productive at all levels of investment (thus allowing for unbounded expansion of output and consumption).

In their paper, Mitra and Roy (2017) provide an example to show why their result may not hold if the technology is globally productive; in an economy with a deterministic linear production function where the average productivity is always greater than one, they show that there is a continuous and co-monotone Ramsey–Euler policy function that is not optimal.

In this paper, we provide an example of an economy with a deterministic linear production function that is globally *unproductive* i.e., the average productivity is always less than one; we explicitly derive a non-linear solution to the Ramsey–Euler functional equation that is not optimal; this non-optimal consumption function is smooth, strictly convex and strictly increasing in output; also, investment is strictly increasing in output. Together, these two examples show that a continuous and co-monotone Ramsey–Euler policy function need not be optimal once we allow for production functions that do not exhibit bounded growth or are not productive near zero.

⁴ For certain versions of our model, in checking for optimality of a Ramsey–Euler path (from an arbitrary initial stock) the transversality condition may be replaced by an infinite number of "period by period" conditions; see, Brock and Majumdar (1988), Dasgupta and Mitra (1988) and Nyarko (1988). Like the transversality condition, these period-by-period conditions taken together involve the entire stochastic process of consumption and capital and establishing optimality by showing that all of them hold can be difficult to implement.

⁵ Mitra and Roy (2017) illustrate this difficulty through examples.

In many macroeconomic applications⁶, the structure of technology shocks is such that the net return on investment is always negative under adverse realizations of the shock. On the other hand, understanding sustained or long run economic growth requires analysis of models with production technologies that are productive at all levels of investment.⁷,⁸ It is important to understand the nature of conditions on a Ramsey–Euler policy function that can ensure it is optimal when we allow for such production technologies. The key contribution of this paper is that we develop easily verifiable conditions for optimality of Ramsey–Euler policy functions in a more general version of the one sector growth model than that considered in Mitra and Roy (2017); in particular, we do not require the production function to exhibit bounded growth or to be productive near zero.

In each of the two examples mentioned above, the non-optimal Ramsey–Euler policy function is such that the propensity to consume can be arbitrarily small. We show that some restriction on the behavior of the propensity to consume can play an important role in ensuring optimality of a Ramsey–Euler policy function.

For production functions where the random shock enters multiplicatively, we show that a Ramsey–Euler policy function is optimal if (i) it is either continuous or comonotone, and (ii) the propensity to consume is bounded away from zero; condition (ii) is required to hold only if the worst case production function is unproductive near zero or if the best case production allows for unbounded expansion. This result is a generalization of the optimality conditions in Mitra and Roy (2017). Note that production functions with multiplicative shock structure are widely used in macroeconomics and resource economics; further, the deterministic production function can be seen as a special case of multiplicative shock.

For more general production functions where the random shock is not necessarily multiplicative, we show that a Ramsey–Euler policy is optimal if (i) holds and the propensity to consume is bounded below by a generalized lower bound that depends on the extent of variation (due to random shock) in the elasticity of the production function.

We also show that if the utility function belongs to a special family (that includes, for instance, all bounded utility functions), then conditions (i) and (ii) mentioned above continue to be sufficient for optimality of a Ramsey–Euler policy even if the random shock is not multiplicative.

It is well known that in our model, the optimal consumption policy function is unique, continuous, and both the optimal consumption and investment are nondecreasing (in fact, strictly increasing) in current output; further, if the optimal policy is interior it must satisfy the Ramsey–Euler condition.⁹ This paper shows that these

🖉 Springer

⁶ Similarly, renewable resources stocks may not be able to regenerate and grow (regardless of the stock size and the amount of harvesting) if environmental conditions are highly adverse.

⁷ For analysis of exogenous growth models where the technology may be "productive at infinity" see, among many others, Gale and Sutherland (1968), Levhari and Srinivasan (1969), Majumdar and Zilcha (1987), Jones and Manuelli (1990), de Hek (1999) and de Hek and Roy (2001).

⁸ See, for instance, Jones et al. (2005). In many applcations, the shocks enter the production function multiplicatively and are assumed to have a lognormal distribution. Our framework however assumes that the shocks are bounded.

⁹ See, for instance, Kamihigashi (2007).

global properties of the policy function that are necessary for optimality may not be sufficient for optimality once we allow for production technologies that are potentially unproductive at zero or productive at infinity. Optimality is however ensured if one can, in addition, verify a condition on the limiting behavior of the propensity to consume (though this may not be necessary for optimality); taken together, they replace the transversality condition in the set of sufficient conditions for the optimality.

Continuity or monotonicity of the Ramsey–Euler policy can be easily verified; it is also easy to verify whether our condition on the propensity to consume is satisfied (for instance, whether it is bounded away from zero) for a candidate consumption function. Our result allows us to immediately verify optimality of explicit solutions to the Euler equation in certain applications with specific functional forms for the utility and production functions where the policy function is linear so that propensity to consume is constant.¹⁰ Linearity is however an exception, rather than the rule. As new examples are developed in the future with non-linear Ramsey–Euler consumption functions (as in the examples outlined in this paper), our result will continue to be useful as a way to verify optimality.¹¹ Our main result can also be a useful theoretical tool in proving optimality of an implicitly defined policy function.

The paper is organized as follows. Section 2 outlines the model, the assumptions and some definitions. Section 3 outlines some benchmark results for the classical version of the model with bounded growth technology that is productive near zero. Section 4 outlines two important examples to illustrate the fact that a continuous and co-monotone Ramsey–Euler consumption function may not be optimal and that the main result in Mitra and Roy (2017) may not hold for a more general class of production technologies. Section 5 contains the main results of the paper on sufficient conditions for optimality of a Ramsey–Euler policy. Section 6 concludes. Section 7 is the appendix and contains proofs of all results and some details of the example in Sect. 4.1 (including an explanation of the method by which we arrived at an explicit nonlinear solution to the Ramsey–Euler functional equation).

2 The model

We consider an infinite horizon one-good representative agent economy. Let \mathbb{R}_+ (\mathbb{R}_{++}) denote the set of all non-negative (strictly positive) real numbers. Time is discrete and is indexed by t = 0, 1, 2, ... At each date $t \ge 0$, the representative agent observes the current stock of output $y_t \in \mathbb{R}_+$ and chooses the level of current investment x_t , and

¹⁰ See, for instance, Benhabib and Rustichini (1994).

¹¹ Our alternative sufficient condition for optimality of a Ramsey–Euler policy is based on the duality approach to the characterization of optimality. A different approach, based on dynamic programming, involves guessing the value function from the Ramsey–Euler condition and verifying that this "candidate" value function satisfies the the Bellman equation (see, for instance, Stokey and Lucas 1989). This approach is useful if the solution to the Bellman equation is unique (for instance, if the utility function is bounded in the stochastic growth model). Recent advances have extended the applicability of this approach to unbounded utility functions; see, among others, Rincón-Zapatero and Rodriguez-Palmero (2003), Matkowski and Nowak (2011) and Kamihigashi (2014).

the current consumption level c_t , such that

$$c_t \ge 0, x_t \ge 0, c_t + x_t \le y_t.$$

This generates y_{t+1} , the output stock next period through the relation

$$y_{t+1} = f(x_t, r_{t+1})$$

where f(x, r) is the production function and r_{t+1} is a random production shock realized at the beginning of period (t + 1).

2.1 Production

We now describe aspects of the above mentioned production process formally. We begin by specifying the nature of the exogenous shocks to production as follows

(**R.1**) The sequence of random shocks $\{r_t\}_{t=1}^{\infty}$ is assumed to be an independent and identically distributed random process defined on a probability space (Ω, \mathcal{F}, P) , where the marginal distribution is denoted by μ . The support of this distribution function is a non-empty compact set $A \subset \mathbb{R}$. The distribution function corresponding to μ is denoted by F.

The *production function* is a map f from $\mathbb{R}_+ \times A$ to \mathbb{R}_+ . We impose the following assumptions¹² on the production function f:

(T.1) Given any $r \in A$, f(., r) is assumed to be continuous, strictly increasing and concave on \mathbb{R}_+ , with f(0, r) = 0, and differentiable on \mathbb{R}_{++} , with $f'(\cdot, r) > 0$ on \mathbb{R}_{++} . Further, for any $x \ge 0$, $f(x, .) : A \to \mathbb{R}_+$, is a (Borel) measurable function.

Define the lower envelope production function $f(x) : \mathbb{R}_+ \to \mathbb{R}_+$ by

$$\underline{f}(x) = \inf_{r \in A} f(x, r).$$

It is easy to check that $\underline{f}(x)$ is non-decreasing on \mathbb{R}_+ and $\underline{f}(0) = 0$. Further, $\underline{f}(x)$ is concave on \mathbb{R}_+ . It follows that the "worst case" average productivity of investment $[\underline{f}(x)/x]$ is non-increasing in x on \mathbb{R}_{++} . The upper envelope production function $\overline{f}(x)$ is defined on \mathbb{R}_+ by:

$$\overline{f}(x) = \sup_{r \in A} f(x, r)$$

We assume that:

(**T.2**)

$$\underline{f}(x) > 0, \, \overline{f}(x) < \infty \text{ for all } x > 0.$$

¹² Note that we do not require the production function to be monotonic or continuous in the realization of the production shocks.

Given an initial stock $y \ge 0$, a stochastic process $\{y_t(y, \omega), c_t(y, \omega), x_t(y, \omega)\}$ is *feasible* from y if it satisfies $y_0 = y$, and:

(i)
$$c_t(y, \omega) \ge 0, x_t(y, \omega) \ge 0$$
 for $t \ge 0$
(ii) $c_t(y, \omega) + x_t(y, \omega) \le y_t(y, \omega), y_{t+1}(y, \omega) = f(x_t(y, \omega), r_{t+1}(\omega))$ for $t \ge 0$

and if for each $t \ge 0$ { $c_t(y, \omega), x_t(y, \omega)$ } are \mathcal{F}_t adapted where \mathcal{F}_t is the (sub) σ -field generated by partial history from periods 0 through t.¹³

2.2 Preferences

Consumption in each period generates an immediate return according to a utility function, $u : \mathbb{R}_{++} \to \mathbb{R}$. The following assumption is imposed on the utility function:

(U.1) u is continuously differentiable, strictly increasing and strictly concave on \mathbb{R}_{++} with u' > 0 on \mathbb{R}_{++} .

We define

$$u(0) \equiv \lim_{c \downarrow 0} u(c),$$

where the limit is allowed to be finite or $-\infty$.

The agent discounts future utility using a time invariant discount factor denoted by $\rho \in (0, 1)$.

2.3 The optimization problem

Given initial stock $y \ge 0$, the representative agent's objective is to maximize the expected value of the discounted sum of utilities from consumption:

$$E\left[\sum_{t=0}^{\infty}\rho^{t}u(c_{t})\right]$$

subject to feasibility constraints.

Given $y \ge 0$, define the stochastic process of consumption $\{c_t^M(y,\omega)\}$ by: $c_0^M(y,\omega) = y, c_{t+1}^M(y,\omega) = f(c_t^M(y,\omega), r_{t+1}(\omega))$ for all $t \ge 0$. Then, for every ω and $t, c_t^M(y,\omega)$ is an upper bound on feasible consumption in period t. We assume that:

(D.1) For all $y \ge 0$,

$$E\left[\sum_{t=0}^{\infty}\rho^{t}u(c_{t}^{M}(y,\omega))_{+}\right]<\infty$$

where $u(c)_{+} = \max\{u(c), 0\}.$

¹³ We skip formal definitions of sigma fields and sub sigma fields as these constructs are standard in the theory of stochastic processes.

Assumption (**D.1**) ensures that for any feasible stochastic process $\{y_t(y, \omega), c_t(y, \omega), x_t(y, \omega)\}$ from $y \ge 0$, the objective of the representative agent

$$E\left[\sum_{t=0}^{\infty}\rho^{t}u(c_{t}(y,\omega))\right]$$

is well defined though it may equal $-\infty$, and that (see, Kamihigashi 2007)

$$E\left[\sum_{t=0}^{\infty}\rho^{t}u(c_{t}(y,\omega))\right] = \sum_{t=0}^{\infty}\rho^{t}E[u(c_{t}(y,\omega))]$$
(1)

Note that (**D.1**) is always satisfied if either *u* is bounded above or alternatively, if $\limsup_{x\to\infty} [\overline{f}(x)/x] < 1$ i.e., the technology exhibits bounded growth.

Given initial stock $\overline{y} \ge 0$, a feasible stochastic process $\{y_t(\overline{y}, \omega), c_t(\overline{y}, \omega), x_t(\overline{y}, \omega)\}$ is *optimal* from \overline{y} if for every feasible stochastic process $\{y'_t(\overline{y}, \omega), c'_t(\overline{y}, \omega), x'_t(\overline{y}, \omega)\}$ from \overline{y} ,

$$E\left[\sum_{t=0}^{\infty} \rho^{t} u(c_{t}(\overline{y}, \omega))\right] \ge E\left[\sum_{t=0}^{\infty} \rho^{t} u(c_{t}'(\overline{y}, \omega))\right]$$

2.4 The optimal consumption function

A consumption (policy) function, is a function $c : \mathbb{R}_+ \to \mathbb{R}_+$, satisfying:

$$0 \le c(y) \le y$$
 for all $y \in \mathbb{R}_+$

Note that this implies c(0) = 0. Associated with a consumption function $c(\cdot)$, is an *investment (policy) function* $x : \mathbb{R}_+ \to \mathbb{R}$, defined by

$$x(y) = y - c(y)$$
 for all $y \in \mathbb{R}_+$

Thus, the investment function $x(\cdot)$ satisfies:

$$0 \le x(y) \le y$$
 for all $y \in \mathbb{R}_+$

A feasible stochastic process $\{y_t(\overline{y}, \omega), c_t(\overline{y}, \omega), x_t(\overline{y}, \omega)\}$ is said to be *generated by* a consumption function c(y) from initial stock $\overline{y} \in \mathbb{R}_+$ if for all $\omega \in \Omega$

$$y_0(\overline{y},\omega) = \overline{y}; \ y_{t+1}(\overline{y},\omega) = f(y_t(\overline{y},\omega) - c(y_t(\overline{y},\omega)), r_{t+1}(\omega)) \text{ for } t \ge 0;$$

$$c_t(\overline{y},\omega) = c(y_t(\overline{y},\omega)), \ x_t(\overline{y},\omega) = x(y_t(\overline{y},\omega)) = y_t(\overline{y},\omega) - c(y_t(\overline{y},\omega)) \text{ for } t \ge 0.$$

A consumption function c(y) is called an *optimal* consumption function if for every $\overline{y} \in \mathbb{R}_+$, the feasible stochastic process $\{y_t(\overline{y}, \omega), c_t(\overline{y}, \omega), x_t(\overline{y}, \omega)\}$ generated by c(y) is optimal from initial stock \overline{y} . A consumption function c(y) is said to be *interior* (or, to satisfy *interiority*) if

$$0 < c(y) < y$$
 for all $y > 0$.

A consumption function c(y) is said to be *co-monotone* if c(y) and x(y) = y - c(y) are non-decreasing in y on \mathbb{R}_+ .

2.5 Ramsey–Euler and transversality conditions

An *interior* consumption function c(y) is said to satisfy the Ramsey–Euler condition if

$$u'(c(y)) = \rho \int_{A} u'(c(f(y - c(y), r))) f'(y - c(y), r) dF(r) \text{ for all } y > 0 \quad (\text{RE})$$

In this case we refer to the consumption function c(y) as a *Ramsey–Euler* (consumption) *policy*.

For any *interior* consumption function c(y), the feasible stochastic process $\{y_t(\overline{y}, \omega), c_t(\overline{y}, \omega), x_t(\overline{y}, \omega)\}$ generated by the consumption function c(y) from any initial stock $\overline{y} > 0$ satisfies:

$$y_t(\overline{y}, \omega) > 0, c_t(\overline{y}, \omega) > 0, x_t(\overline{y}, \omega) > 0$$
 for all $t \ge 0$ and for all $\omega \in \Omega$.

An interior consumption function c(y) is said to satisfy the *transversality condition* if for all $\overline{y} > 0$:

$$\lim_{t \to \infty} E\{\rho^t u'(c_t(\overline{y}, \omega))x_t(\overline{y}, \omega)\} = 0$$
 (TC)

where { $y_t(\overline{y}, \omega), c_t(\overline{y}, \omega), x_t(\overline{y}, \omega)$ } is the feasible stochastic process generated by the consumption function c(y) from initial stock \overline{y} .

3 Optimality of Ramsey–Euler policy: benchmark

It is known that if a consumption function is interior, satisfies the Ramsey–Euler condition (RE) and the transversality condition (TC), then it is an optimal consumption function; in other words, a Ramsey–Euler policy is optimal if it satisfies the transversality condition (TC). This was established by Mirman and Zilcha (1975) in the "bounded growth" case; it has since been established in more general settings. A specific version of this sufficiency result (for the model outlined in Sect. 2) is reported in this paper as Lemma 1 (in the Appendix) and is used in the proof of our main results. It should be mentioned that the transversality condition (TC) has also been shown to be necessary for optimality of a Ramsey–Euler policy.

As mentioned earlier, the transversality condition essentially involves the entire stochastic process of consumption and capital generated by a policy function; it cannot be verified immediately by inspecting the policy function. Depending on the specific utility and production functions, verification of the transversality condition can require some work (see, Mitra and Roy 2017 for some more discussion of this issue). It is therefore interesting to explore whether there are alternative conditions for optimality of a policy function that are easier to verify than the transversality condition; in other words, can the transversality condition be replaced by some fairly apparent properties of the policy function.

Mitra and Roy (2017) establish alternative conditions for optimality of a policy function for the "canonical" version of the one sector optimal stochastic growth model where the technology exhibits bounded growth (i.e., there is a maximum sustainable capital stock) and is productive (for sure) near zero. The main result in that paper is stated below for ease of comparison:

Proposition 1 (Theorem 1, Mitra and Roy 2017) Assume the following:

(E.1) There is K > 0 such that $\left[\overline{f}(x)/x\right] < 1$ for all x > K(E.2) $\lim_{x \downarrow 0} [f(x)/x] > 1$

Suppose that $c(\cdot)$ is an interior consumption function. Then the following statements are equivalent:

- (a) c(y) is continuous and satisfies the Ramsey–Euler condition (RE)
- (b) c(y) and y c(y) are nondecreasing on \mathbb{R}_+ (i.e., c(y) is co-monotone) and c(y) satisfies the Ramsey–Euler condition (RE)
- (c) c(y) and y-c(y) are strictly increasing on \mathbb{R}_+ and c(y) satisfies the Ramsey–Euler condition (RE)
- (d) c(y) is optimal.

The key implication of this result is that for the canonical version of the model, a Ramsey–Euler policy function is optimal as long as it is continuous (or alternatively, co-monotone).

The proof of this result in Mitra and Roy (2017) uses the end-point conditions (E.1) and (E.2) on the production technology. This naturally leads to the question whether their result extends to economic environments where either (E.1) or (E.2) does not hold i.e., the production technology is not necessarily productive near zero for all realizations of the shock or alternatively, allows for unbounded expansion of capital and consumption (or both). In the next section, we outline two examples to show that their result may not hold if the production technology does *not* satisfy either condition (E.1) or condition (E.2).

4 Non-optimal continuous and co-monotone Ramsey–Euler policy: two examples

In this section, we outline two examples of economies with *deterministic* production technologies that do not satisfy the endpoint conditions assumed in Proposition 1. In first example, the production function is unproductive at all levels of investment and therefore violates condition (E.2). In the second example, the production function is productive at all levels of investment i.e., allows for unbounded expansion of

capital and consumption, and therefore violates condition (E.1). For each example, we explicitly specify an interior consumption function that solves the Ramsey–Euler equation (RE), is continuous and co-monotone, but is not optimal.

4.1 Example 1: unproductive technology

In this example, the production function is deterministic, linear and unproductive at all positive input levels. It is given by

$$f(x) = \frac{x}{2} \text{ for all } x \ge 0 \tag{2}$$

We specify the utility function *u* to be:

$$u(c) = \begin{cases} \ln c & \text{for all } c > 0 \\ -\infty & \text{for } c = 0 \end{cases}$$
(3)

Finally, let the discount factor $\rho = \frac{1}{2}$. Then, all of our assumptions in Sect. 2 are satisfied.

The Ramsey-Euler functional equation (RE) for this example reduces to:

$$c\left(\frac{y-c(y)}{2}\right) = \frac{c(y)}{4} \text{ for all } y > 0$$
(4)

It is easy to see that the consumption function:

$$c^*(y) = \frac{y}{2}$$
 for all $y \ge 0$

solves the Ramsey–Euler function equation (4) and the path $\{c_t^*, x_t^*, y_t^*\}$ generated by this policy function satisfies the transversality condition (TC).¹⁴Therefore, (using for instance, Lemma 1 in the appendix), $c^*(y)$ is in fact the optimal consumption policy function. Note that strict concavity of the utility and production functions implies that the optimal consumption function is unique.

We now show that there is a continuous and co-monotone solution to the Ramsey– Euler functional equation (4) that is different from $c^*(y)$ and is therefore, not optimal. Consider the function $\phi(y)$ defined by:

$$\phi(y) = \frac{(1+4y) - (1+8y)^{\frac{1}{2}}}{8} \text{ for all } y \ge 0$$
(5)

Note that $\phi(0) = 0$, and since $(1 + 8y)^{\frac{1}{2}} < (1 + 8y + 16y^2)^{\frac{1}{2}} = (1 + 4y)$ for all y > 0, we have $\phi(y) > 0$ for all y > 0. Further, since $(1 + 8y)^{\frac{1}{2}} > 1$ for all y > 0, we have:

$$\phi(y) < (4y/8) = (y/2) = c^*(y) \text{ for all } y > 0$$
 (6)

```
\overline{{}^{14} \ \rho^t u'(c_t^*))} x_t^* = (1/2)^t \to 0 \text{ as } t \to \infty.
```

Thus, ϕ is an interior consumption function. Clearly, ϕ is continuous and differentiable on \mathbb{R}_+ . By differentiating (5), we see that:

$$8\phi'(y) = 4 - \frac{4}{(1+8y)^{\frac{1}{2}}} > 0$$
 for all $y > 0$

so that $\phi'(y) > 0$ for y > 0, and ϕ is strictly increasing on \mathbb{R}_+ . Further, $\phi'(y) < \frac{1}{2}$ for all y > 0. Thus, the interior consumption function $\phi(y)$ is continuous and comonotone on \mathbb{R}_+ . We now claim that $c(y) = \phi(y)$ is a solution to the Ramsey–Euler functional equation (4). To see this, define $\psi : \mathbb{R}_+ \to \mathbb{R}_+$ by:

$$\psi(c) = 2c + c^{\frac{1}{2}} \text{ for all } c \ge 0$$
(7)

Note that $\psi(0) = 0$, and $\psi(c) > 0$ for all c > 0. In fact, $\psi(c) > 2c$ for all c > 0. Further, $\psi(c)$ is strictly increasing and strictly concave in c on \mathbb{R}_+ . One can directly verify that the functions ψ and ϕ are inverses of each other i.e., $\psi(\phi(y)) = y$ for all $y \ge 0$ and $\phi(\psi(c)) = c$ for all $c \ge 0$; details are contained in Sect. 7.1 of the Appendix.

The difficulty in solving the functional equation (4) arises from the composition of the unknown function with itself on the left-hand side. To get around this difficulty, one writes down its conjugate functional equation:

$$g(c/4) = (1/2)(g(c) - c) \text{ for } c \ge 0$$
 (8)

(8) can be rewritten as:

$$g(c) = c + 2g(c/4) \text{ for } c \ge 0$$
 (9)

We show that $g(c) = \psi(c)$, where ψ is defined by (7), solves (9) and that its inverse $\phi(y)$, defined in (5), is a solution to the Ramsey–Euler functional equation (4). These results are explicitly established in Sect. 7.1 of the Appendix.

One interesting feature of the Ramsey–Euler consumption function $\phi(y)$ in the above example is that the propensity to consume $[\phi(y)/y] \rightarrow 0$ as $y \rightarrow 0$. We will see that this is a possible source of non-optimality of $\phi(y)$.

To the best of our knowledge, this is the first explicit example of a smooth, nonlinear and strictly increasing consumption function that solves the Ramsey–Euler equation in the canonical stochastic growth model. Even though it is not an optimal consumption function, it may be useful for researchers in the field to understand how one would "guess at" a solution like (5) to the Ramsey–Euler functional equation. Section 7.2 in the appendix explains this in details for a somewhat more general set of parameters.

4.2 Example 2: unbounded growth technology

We now outline an example of an economy where the production technology allows for unbounded expansion of consumption and output i.e., the end point condition (E.1) in Proposition 1 does not hold. In this economy, there is a non-optimal Ramsey–Euler

consumption function that is continuous and co-monotone. The example is contained in Mitra and Roy (2017: Example 3, Sect. 5); key aspects are reproduced below for ease of reference. Define the utility function u to be:

$$u(c) = \frac{\sqrt{c}}{1+\sqrt{c}}$$
 for all $c \ge 0$

Then, u satisfies (U.1). The production technology is deterministic and is given by

$$f(x) = 2x$$

which satisfies (T.1). Set $\rho = (1/2)$. Consider the consumption function defined by:

$$c(y) = \begin{cases} (1/2)y \text{ for } 0 \le y \le 2\\ 1 & \text{for } y > 2 \end{cases}$$

Observe that c(y) is interior and continuous; further, c(y) and y - c(y) are nondecreasing in y. For $0 < y \le 2$, we have c(y) = (1/2)y, and f(y - c(y)) = 2(y - (1/2)y) = y, so that c(f(y - c(y))) = (1/2)y = c(y). Thus

$$\rho u'(c\{f(y-c(y))\})f'(y-c(y)) = \frac{1}{2}u'((1/2)y)2 = u'(c(y))$$

verifying (RE) for $y \in (0, 2]$. For y > 2, we have 2(y - 1) = 2y - 2 > 2, and so $c\{f(y - c(y))\} = c\{2(y - 1)\} = 1$. Thus,

$$\rho u'(c\{f(y - c(y))\})f'(y - c(y)) = (1/2)u'(c\{f(y - c(y))\})2$$
$$= u'(1) = u'(c(y))$$

verifying (RE) for y > 2. Finally, consider a different consumption function $\gamma(\cdot)$ defined by:

$$\gamma(y) = (1/2)y$$
 for all $y \ge 0$

Starting from y = 4, the consumption function $\gamma(\cdot)$ generates a path $(\tilde{y}_t, \tilde{c}_t, \tilde{x}_t)$ where consumption $\tilde{c}_t = 2$ for all $t \ge 0$. On the other hand, the path (y_t, c_t, x_t) starting from y = 4, generated by the consumption function $c(\cdot)$, has $y_t \ge 4$ for all $t \ge 0$ and so $c_t = 1$ for all $t \ge 0$, so that the discounted sum of utilities along the path (y_t, c_t, x_t) is strictly smaller than along the path $(\tilde{y}_t, \tilde{c}_t, \tilde{x}_t)$. Thus, $c(\cdot)$ is *not* an optimal consumption function. This concludes the example.

Observe that somewhat similarly to Example 1, an interesting feature of the nonoptimal Ramsey–Euler consumption function c(y) in Example 2 is that the propensity to consume $[c(y)/y] \rightarrow 0$ as $y \rightarrow \infty$.

5 Optimality of Ramsey–Euler policy: sufficient conditions

In this section, we outline properties of a Ramsey–Euler consumption function that are sufficient to ensure that it is optimal even if the production technology is unproductive or allows for unbounded expansion of capital and consumption.

Recall that \overline{f} , \underline{f} are the upper and lower envelopes of the production function defined in Sect. 2; they correspond to "best" and "worst" possible realizations of the random shock.

Define $\overline{K} \ge 0$ by:

$$\overline{K} = \sup\{x \ge 0 : \overline{f}(x) \ge x\}$$

 $\overline{K} = \infty$ if the production technology allows for unbounded growth i.e., $\overline{f}(x) > x$ for all x > 0; further, $\overline{K} = 0$ if the technology is unproductive for sure and $\overline{f}(x) < x$ for all x > 0.

Define $\underline{K} \ge 0$ by:

$$\underline{K} = \inf\{x > 0 : \underline{f}(x) \le x\}$$
$$= \infty, \text{ if } f(x) > x \text{ for all } x > 0$$

 $\underline{K} > 0$ if the "worst case" technology is productive near zero i.e., $\lim_{x \downarrow 0} [\underline{f}(x)/x] > 1$; $\underline{K} = 0$ if $\lim_{x \downarrow 0} [\underline{f}(x)/x] \le 1$ so that¹⁵ $\underline{f}(x) \le x$ for all $x \ge 0$ i.e., the "worst case" technology is globally unproductive. Finally, note that $\underline{K} \le \overline{K}$.

5.1 Main result

In this subsection, we consider the general model outlined in Sect. 2. For each x > 0, $r \in A$, let the inverse elasticity of the production function $\eta(x, r) > 1$ be defined by

$$\eta(x,r) = \frac{f(x,r)}{f'(x,r)x}$$

and let

$$\overline{\eta}(x) = \sup_{r \in A} \eta(x, r), \, \underline{\eta}(x) = \inf_{r \in A} \eta(x, r).$$

We now specify a technical assumption on the production function that is used in the next proposition and simplifies our analysis considerably:

(**T.3**) (i) There exists $a, b \in A, z_0, z_1 \in \mathbb{R}_{++}$ such that

$$\frac{f(x) = f(x, a) \text{ for all } x \in [0, z_1]}{\overline{f}(x) = f(x, b) \text{ for all } x \ge z_2}$$

¹⁵ Note that f is concave on \mathbb{R}_+ .

(ii)

$$\tau_0 = \lim_{x \to 0} \sup \frac{\overline{\eta}(x)}{\eta(x, a)} < \infty$$

$$\tau_\infty = \lim_{x \to \infty} \sup \frac{\overline{\eta}(x)}{\eta(x, b)} < \infty$$

Under (**T.3**)(i), there is a specific "worst" case production shock *a* associated with the lower envelope of the production function for investment levels close to zero, and a specific "best" case production shock *b* associated with the upper envelope of the production function when investment is large enough. (**T.3**)(ii) essentially requires that the variation due to random shock in the elasticity of the production function at zero and infinity are bounded. If the random shock enters the production function multiplicatively, then $\eta(x, r)$ is independent of r so that $\tau_0 = \tau_{\infty} = 1$ and assumption (**T.3**)(ii) is satisfied. Note that (**T.3**) is also satisfied by some well known production functions that are not ordered by the shock such as $f(x, r) = x^r$, $r \in A \subset (0, 1)$.

Recall that $f'(0, r) = \lim_{x\to 0^+} f'(x, r)$ is the marginal productivity at zero for realization *r* of the random shock $(f'(0, r) \text{ may equal } +\infty)$.

We are now ready to state our main proposition:

Proposition 2 Assume (T.3). Consider a Ramsey–Euler consumption function c(y) that is either continuous or co-monotone on \mathbb{R}_+ . Further, suppose that

$$\lim_{y \to 0} \inf \frac{c(y)}{y} > 1 - \frac{1}{\tau_0}, \text{ if } f'(0, a) \le \tau_0 \tag{GP1}$$

$$\lim_{y \to \infty} \inf \frac{c(y)}{y} > 1 - \frac{1}{\tau_{\infty}}, \ if \overline{K} = \infty$$
(GP2)

Then, c(y) is optimal (and is, in fact, the unique optimal consumption function)

Proposition 2 provides a set of verifiable properties of a Ramsey–Euler policy function that ensures it is optimal in environments that allow for unproductive technology as well as unbounded growth.

The proof of Proposition 2 is based on showing that the transversality condition (TC) holds i.e., $\rho^t E\{u'(c_t)x_t\} \rightarrow 0$ as $t \rightarrow \infty$ where $\{c_t\}$ and $\{x_t\}$ are the consumption and investment processes generated by the continuous (and co-monotone) Ramsey–Euler consumption function c(y). This is trivial if the corresponding output process $\{y_t\}$ lies almost surely in a closed interval that is bounded away from zero. The difficulty arises when output and consumption are not bounded away from zero or infinity with positive probability. Our proof is based on using the fact that $x_t \leq y_t$ with probability one and showing that each term of the sequence $\{\rho^t E(u'(c_t)y_t)\}_{t=0}^{\infty}$ is a *contraction* of its previous term. This is different from the proof of optimality of Ramsey–Euler policy in Mitra and Roy (2017) where the transversality condition is shown to hold without demonstrating such a contraction property; the arguments in that proof cannot be easily extended to production functions that are unproductive near zero or productive at infinity.

The sufficient conditions for optimality in Mitra and Roy (2017) impose no restriction on the propensity to consume. The two examples in the previous section indicate that some restrictions on the propensity to consume are needed for a Ramsey–Euler policy function to be optimal when the technology is unproductive or allows for unbounded growth. Conditions (GP1) and (GP2) impose lower bounds on the propensity to consume; the bounds depend on the extent of variation in the elasticity of the production function due to random shocks. These are sufficient conditions; we are unable to determine whether they are necessary for optimality in such technological environments.

It is worth noting that if the production function satisfies bounded growth i.e., $\overline{K} < \infty$, then condition (GP2) no longer applies. However, condition (GP1) may continue to apply even if the production technology is productive near zero i.e., $\underline{K} > 0$. Thus, the sufficient conditions in Proposition 2 are potentially stronger than and do not reduce to the optimality conditions in Mitra and Roy (2017) for production functions that satisfy the assumptions in that paper (or alternatively, Proposition 1 under assumptions E.1 and E.2).

Proposition 2 also yields the following simpler result:

Corollary 1 Consider a Ramsey–Euler consumption function c(y) that is continuous or co-monotone on \mathbb{R}_+ . Further, suppose that

$$\tau = \sup_{x>0} \frac{\overline{\eta}(x)}{\eta(x)} < \infty \text{ and } \inf_{y>0} \frac{c(y)}{y} > 1 - \frac{1}{\tau}$$

Then, c(y) is optimal and in fact, is the unique optimal consumption function.

5.2 Multiplicative shock

In this subsection, we consider production functions where the random shock is multiplicative. Such production functions are widely used in the literature; further, both examples in the previous section deal with deterministic production functions that can be viewed as special cases of multiplicative shock. In particular, for this subsection we assume:

$$f(x,r) = q(r)h(x), r \in A, x \ge 0$$
(10)

Assumptions (T.1), (T.2) and (T.3) on f(x, r) hold under the following restrictions on the function h and q:

(M.1) $h : \mathbb{R}_+ \to \mathbb{R}_+$ is continuous, concave and strictly increasing, h(0) = 0 and h is differentiable on \mathbb{R}_{++} , h'(x) > 0 for all x > 0 and $h'(0) = \lim_{x \to 0} h'(x) \in \mathbb{R}_+ \cup \{+\infty\}$ satisfies

(M.2) $q: A \to \mathbb{R}_{++}$ is Borel-measurable and there exists $a, b \in A$ such that

$$q(a) \le q(r) \le q(b)$$
 for all $r \in A$.

Once again, *a* and *b* are respectively the worst and best shocks. Note that if q(r) = 1 for all *r*, we have a deterministic production function where f(x, r) = h(x).

It is easy to check that

$$\eta(x,r) = \frac{f(x,r)}{f'(x,r)x} = \frac{h(x)}{h'(x)x}$$

is independent of r so that

$$\tau_0 = \tau_\infty = 1.$$

Also, observe that $\underline{K} = 0$ if, and only if,

$$q(a)h'(0) \le 1 = \tau_0.$$

Proposition 2 therefore immediately yields:

Corollary 2 Consider the class of production functions where the random shock is multiplicative and in particular, (10) holds under restrictions (M.1) and (M.2). Let c(y) be a Ramsey–Euler consumption function that is either continuous or co-monotone on \mathbb{R}_+ . Further, suppose that the propensity to consume (c(y)/y) satisfies:

$$\lim_{y \to 0} \inf \frac{c(y)}{y} > 0, \ if \underline{K} = 0 \tag{C.1}$$

$$\lim_{y \to \infty} \inf \frac{c(y)}{y} > 0, \ if \overline{K} = \infty$$
(C.2)

Then, c(y) is optimal.

Note that under the assumptions of Mitra and Roy (2017), $\underline{K} > 0$ and $K < \infty$ so that condition C.1 and C.2 in Proposition 2 do not apply and continuity or comonotonicity of Ramsey–Euler policy is sufficient for optimality. In other words, for the multiplicative shock case, the sufficient conditions for optimality in Proposition 2 reduce to the optimality conditions in Mitra and Roy (2017) under their assumptions (or alternatively, to those in Proposition 1 under restrictions E.1 and E.2). Within the class of production functions with multiplicative shocks, Proposition 2 generalizes the sufficient conditions in Mitra and Roy (2017) to a larger set of production functions.

5.3 A special class of utility functions

In this subsection, we restrict attention to a class of utility functions while allowing the production function to have a fairly general structure. We show that for this class of utility functions, a Ramsey–Euler policy is optimal as long as the propensity to consume is bounded away from zero. In particular, we assume that in addition to (U.1) and (U.2), the utility function *u* satisfies:

(U.3) u'(c)c is bounded on \mathbb{R}_{++} ; in particular, there exists $M \in \mathbb{R}_{++}$ such that u'(c)c < M for all c > 0.

Note that (**U.3**) is satisfied if *u* is bounded on \mathbb{R}_+ .¹⁶ Of course, u'(c)c may be be bounded on \mathbb{R}_{++} even if *u* is not bounded (for instance, $u(c) = \ln c$).

Proposition 3 Assume (U.3). Consider a Ramsey–Euler consumption function c(y) that is either continuous or co-monotone on \mathbb{R}_+ ; further,

$$\inf_{y>0}\frac{c(y)}{y}>0.$$

Then, c(y) is optimal.

Note that unlike Propositions 2, 3 does not require any restriction like (**T.3**) on the production function; unlike Proposition 2, the proof of Proposition 3 is not based on a "contraction" argument.

5.4 Application

The sufficient conditions for optimality of Ramsey–Euler policy can be useful in verifying optimality of explicit solutions to the Ramsey–Euler functional equation for specific utility and production functions. For instance, consider a CES utility function:

$$u(c) = \frac{c^{1-\sigma}}{1-\sigma}$$

where $\sigma > 0, \sigma \neq 1$. The production function is given by

$$f(x,r) = rx$$

and $\{r_t\}$ is a sequence of i.i.d. random variable with distribution *F* with support [a, b], $0 < a < b < \infty$. Note that the production technology may be unproductive (at least for certain realizations of the shock) as well as allow for unbounded expansion with positive probability. It is assumed that

$$k = [\rho E(r_t^{1-\sigma})]^{\frac{1}{\sigma}} < 1.$$

It is easy to check (and fairly well known) that the linear consumption function

$$\widetilde{c}(y) = (1-k)y$$

$$b - u(c) \le u(0) - u(c) \le u'(c)(-c)$$
 for all $c > 0$

and so:

$$u'(c)c \le u(c) - b \le B - b$$
 for all $c > 0$

i.e., the function u'(c)c is bounded on \mathbb{R}_{++} .

🖉 Springer

¹⁶ Let $u : \mathbb{R}_+ \to \mathbb{R}$ be a bounded, strictly concave and continuously differentiable function on \mathbb{R}_+ so that in particular $b \le u(c) \le B$ for all $c \ge 0$, for some $b, B \in \mathbb{R}$. Then

solves the Ramsey–Euler equation for this problem. To assert that $\tilde{c}(y)$ is optimal by verifying that the consumption and investment process generated by $\tilde{c}(y)$ satisfies the transversality condition can take some work. On the other hand, by direct appealing to Corollary 2 and noting that the propensity to consume c(y)/y = 1 - k is a strictly positive constant, we can immediately assert optimality of $\tilde{c}(y)$.¹⁷

As new examples are developed in the future with non-linear Ramsey–Euler consumption functions our result will continue to be useful as a way to verify optimality.

6 Conclusion

The classical version of the one sector convex model of stochastic optimal growth assumes that the technology is productive near zero and exhibits bounded growth with probability one. In this framework, it has been shown that a policy function satisfying the Ramsey-Euler condition is optimal as long it is continuous or alternatively, if both consumption and investment are non-decreasing in current output. We outline two counterexamples to show that this result may not hold once the classical model is generalized to accommodate production functions that may be globally unproductive for bad realizations of the shock or allow for unbounded expansion of consumption and output. Our analysis indicates that a probable source of this non-optimality is low propensity to consume exhibited by the candidate policy function. We show that in our more general framework, a Ramsey-Euler policy function is optimal if in addition to continuity or monotonicity properties, we can also verify a condition on the propensity to consume. For production functions with multiplicative shock, our condition simply requires the propensity to consume be bounded away from zero; a generalization of this lower bound is shown to be sufficient for optimality in the case of non-multiplicative shock; weaker conditions are outlined for a restricted class of utility functions that includes bounded utility. The sufficient conditions for optimality outlined in this paper can be significantly easier to verify than the transversality condition.

Our analysis is a step forward in characterizing alternative conditions for optimality in a class of dynamic optimization models that includes the stochastic growth model. It will be useful to extend our analysis to stochastic growth models with "unbounded shocks" (see, Stachurski 2002; Nishimura and Stachurski 2005; Kamihigashi 2007) and irreversible investment (see, Olson 1989).

7 Appendix

7.1 Details of Example 1

(a) ψ and ϕ are inverses of each other:

¹⁷ It is worth noting for this family of utility and production functions and following the steps outlined in Sect. 7.2 ,it is possible to derive non-linear policy functions that satisfy the Ramsey–Euler condition (similar to Example 1 in Sect. 4) but do not satisfy our sufficient conditions for optimality; these non-linear policies are in fact, not optimal.

We first show that $\psi(\phi(y)) = y$ for all $y \ge 0$. To this end, let us note that, by (5), for all $y \ge 0$,

$$\phi(y) = \frac{(1+4y) - (1+8y)^{\frac{1}{2}}}{8} = \left[\frac{(1+8y)^{\frac{1}{2}} - 1}{4}\right]^2$$

which yields:

$$(1+8y) = \{1+4[\phi(y)]^{\frac{1}{2}}\}^2 = 1+16\phi(y)+8[\phi(y)]^{\frac{1}{2}}$$

and this implies:

$$y = 2\phi(y) + [\phi(y)]^{\frac{1}{2}} = \psi(\phi(y))$$

by using (7) and noting that $\phi(y) \ge 0$ for all $y \ge 0$. We now show that $\phi(\psi(c)) = c$ for all $c \ge 0$. We start with the following identity for all $c \ge 0$:

$$[1+4c^{\frac{1}{2}}]^2 = [1+8(2c+c^{\frac{1}{2}})]$$

which can be rewritten as:

$$2 + 8[2c + c^{\frac{1}{2}}] - 2[1 + 8(2c + c^{\frac{1}{2}})]^{\frac{1}{2}} = 16c$$
(11)

.

Using the definition of ψ in (7), we can rewrite (11) as:

$$[2 + 8\psi(c)] - 2[1 + 8\psi(c)]^{\frac{1}{2}} = 16c$$

so that:

$$\frac{[1+4\psi(c)]-[1+8\psi(c)]^{\frac{1}{2}}}{8} = c$$

and using the definition of ϕ in (5), we obtain:

$$\phi(\psi(c)) = \frac{[1+4\psi(c)] - [1+8\psi(c)]^{\frac{1}{2}}}{8} = c.$$

(b) $g(c) = \psi(c)$ is a solution to conjugate functional equation (9) and its inverse $\phi(y)$ is a solution to the Ramsey–Euler functional equation (4).

Let us write:

$$c + 2g(c/4) = c + 2\psi(c/4)$$

= $c + 4(c/4) + 2(c/4)^{\frac{1}{2}}$
= $2c + c^{\frac{1}{2}} = \psi(c) = g(c)$ (12)

where we have used the definition of ψ in the second line and again in the last line of (12). Since we have just demonstrated that ψ is a solution to (9), we can write:

$$\psi(c/4) = \frac{\psi(c) - c}{2} \text{ for } c \ge 0$$
 (13)

Since $\phi(y) \ge 0$ for all $y \ge 0$, we can use (13) to write: $\psi(\phi(y)/4) = \frac{\psi(\phi(y)) - \phi(y)}{2}$ for all $y \ge 0$ i.e.,

$$\psi(\phi(y)/4) = \frac{y - \phi(y)}{2} \quad \text{for all } y \ge 0 \tag{14}$$

Since $\phi(y) \le y$ for all $y \ge 0$, we can apply the function ϕ to both sides of (14) to get: $\phi[\psi(\phi(y)/4)] = \phi\left(\frac{y-\phi(y)}{2}\right)$ for all $y \ge 0$ i.e.,

$$\frac{\phi(y)}{4} = \phi\left(\frac{y - \phi(y)}{2}\right) \text{ for all } y \ge 0$$

so that ϕ solves the Ramsey–Euler functional equation (4).

7.2 Non-linear solution to the Ramsey–Euler functional equation

In the interest of finding nonlinear solutions to the Ramsey–Euler functional equation, it is worthwhile to explain how one arrives at an explicit solution like (5) in Example 1. Consider the deterministic linear production function f(x) = ax where $a \in (0, 1)$. Let the discount factor $\rho \in (0, 1)$. (In Example 1, we chose $a = \rho = 1/2$.) The utility function *u* is as specified in Example 1 i.e., $u(c) = \ln c$. The Ramsey–Euler functional equation is then given by:

$$c(a(y - c(y))) = \rho ac(y) \text{ for all } y > 0$$
(15)

In what follows that our aim is to find *some* solution of (15) which is distinct from the well-known linear solution: $c(y) = (1 - \rho)y$ for all $y \ge 0$. We can confine our search to a more restrictive class of solutions [for example, all *continuous and strictly increasing* solutions to the functional equation (15)].

Step 1: [The Conjugate Functional Equation]

As already noted in the text, the difficulty in solving the functional equation (15) arises from the composition of the unknown function with itself on the left-hand side. To get around this difficulty, one writes down its *conjugate* functional equation:

$$g(\rho ac) = a(g(c) - c) \text{ for } c \ge 0$$
(16)

If $g : \mathbb{R}_+ \to \mathbb{R}_+$ is a function satisfying $g(c) \ge c$ for all $c \ge 0$, which solves the conjugate functional equation (16), and there is a function $h : \mathbb{R}_+ \to \mathbb{R}_+$ such that $h(y) \le y$ for all $y \ge 0$, and:

$$(i) h(g(c)) = c \text{ for all } c \ge 0$$

(ii) $g(h(y)) = y \text{ for all } y \ge 0$ (17)

then *h* solves the functional equation (15). To see this, note that since $h(y) \ge 0$ for all $y \ge 0$, we can use (16) to write: $g(\rho a h(y)) = a[g(h(y)) - h(y)]$ for all $y \ge 0$, and then using (17)(ii) we have:

$$g(\rho a h(y)) = a[y - h(y)] \quad \text{for all } y \ge 0 \tag{18}$$

Since $h(y) \le y$ for all $y \ge 0$, we can apply the function *h* to both sides of (18) to get: $h[g(\rho ah(y))] = h\{a[y - h(y)]\}$ for all $y \ge 0$. Using (17)(i) we then obtain: $\rho ah(y) = h\{a[y - h(y)]\}$ for all $y \ge 0$ so that *h* solves the Ramsey–Euler functional equation (15).

Step 2: [Solving the Conjugate Functional Equation]

Step 1 above suggests that if we can find a solution g to the conjugate functional equation, and g has an inverse (note that g is increasing, in our context), then the inverse of g would solve the Ramsey–Euler functional equation. As we want the solution h(y) to (15) (a consumption function) to be strictly increasing and continuous in y, we should be looking for strictly increasing and continuous solutions g(c) to the conjugate functional equation. Unlike (15), the conjugate functional equation (16) does not involve the composition of g with itself on either side; it belongs to the class of iterated functional equations and can be solved through iterations. Pick any $c \ge 0$, and write, using (16) repeatedly,

$$g(c) = c + (1/a)g(\rho a c)$$

= c + (1/a)[\rho a c + (1/a)g(\rho^2 a^2 c)]
= c + \rho c + (1/a^2)g(\rho^2 a^2 c)

This iteration process can be continued to write for $t \ge 2$,

$$g(c) = [c + \rho c + \rho^2 c + \dots + \rho^t c] + (1/a^{t+1})g(\rho^{t+1}a^{t+1}c)$$
(19)

Note that the term in square brackets in (19) converges to $c/(1 - \rho)$ as $t \to \infty$. So, if the final term in (19) goes to zero as $t \to \infty$, we get $g(c) = c/(1 - \rho)$ for $c \ge 0$, and we arrive at the well-known linear solution to the Ramsey–Euler functional equation: $c(y) = (1 - \rho)y$ for all $y \ge 0$.

But, we don't really know whether the final term in (19) converges to zero as $t \to \infty$. This will depend, after all, on the behavior of g near zero, and g is the unknown function we are trying to find. It is perfectly legitimate to restrict our search for solutions to (16) to a narrow class. Let us, then, confine our search to those g for which the final term in (19) *does* have a limit as $t \to \infty$. The limit will itself be a function of c, and we denote it by q(c); we are interested in functions g for which $q(c) \neq 0$. Then, using (19), and letting $t \to \infty$, we can write:

$$g(c) = [c/(1-\rho)] + q(c) \text{ for all } c \ge 0$$
(20)

The question then arises: what properties must q(c) satisfy in order for (20) to be a valid solution to the conjugate functional equation (16). Given (16) and (20), we must have:

$$(1/a)\{[\rho ac/(1-\rho)] + q(\rho ac)\} = (1/a)g(\rho ac) = (g(c) - c)$$
$$= [c/(1-\rho)] + q(c) - c \text{ for } c \ge 0$$

which can be rewritten as:

$$q(c) = [q(\rho a c)/a] + (1/a) \{ [\rho a c/(1-\rho)] + c - [c/(1-\rho)] = [q(\rho a c)/a]$$
(21)

Recalling Step 1, we want the solution g(c) to (16), given by (20), to be strictly increasing in c, so it is legitimate to confine our search to those functions q(c) which are strictly increasing in c. This would mean that q(c) would be differentiable almost everywhere on \mathbb{R}_+ . So, we might as well confine our search to differentiable and strictly increasing functions q(c).

Differentiating (21) with respect to c, we would get:

$$q'(c) = q'(\rho a c)\rho \tag{22}$$

Using (21) and (22), we obtain (when $q(c) \neq 0$),

$$\frac{q'(c)}{q(c)} = \frac{q'(\rho a c)\rho}{q(\rho a c)/a}$$

which yields:

$$\frac{q'(c)c}{q(c)} = \frac{q'(\rho a c)\rho a c}{q(\rho a c)} \text{ for all } c \text{ such that } q(c) \neq 0$$
(23)

This suggests that we can further restrict our search for q(c) to the iso-elastic class:

$$q(c) = c^{1-\theta} \text{ for all } c \ge 0$$
(24)

where $0 < \theta < 1$ is to be appropriately chosen. That is, our proposed "candidate solution" *g* to the conjugate functional equation is:

$$g(c) = [c/(1-\rho)] + c^{1-\theta}$$
 for all $c \ge 0$ (25)

Note that this has all the desirable properties of g: it is a map from \mathbb{R}_+ to \mathbb{R}_+ , it is continuous and strictly increasing on \mathbb{R}_+ (also differentiable for c > 0), with g(c) > c for all c > 0, and g(0) = 0.

The question then arises: what property must θ satisfy in order for (25) to be a valid solution to the conjugate functional equation (16). Given (16) and (25), we must have:

$$(1/a)\{[\rho ac/(1-\rho)] + (\rho ac)^{1-\theta}\} = (1/a)g(\rho ac) = (g(c) - c)$$

$$= [c/(1-\rho)] + c^{1-\theta} - c \text{ for } c \ge 0$$

That is,

$$(1/a)(\rho a c)^{1-\theta} = c^{1-\theta}$$
 for all $c \ge 0$

This entails the parameter restriction:

$$(1/a)(\rho a)^{1-\theta} = 1 \tag{26}$$

Note that as $\theta \to 0$, the left hand side of (26) converges to $\rho \in (0, 1)$. And, as $\theta \to 1$, the left hand-side of (26) converges to (1/a) > 1. Thus, there is some $\theta \in (0, 1)$, such that (26) holds. We have now verified that:

$$g(c) = [c/(1-\rho)] + c^{1-\theta}$$
 for all $c \ge 0$

with θ chosen to satisfy (26) solves the conjugate functional equation (16).

Step 3: [Solving the Ramsey–Euler Functional Equation (15)]

Note from the form of g in (25) that g is a continuous and strictly increasing function from \mathbb{R}_+ onto \mathbb{R}_+ . Consequently, given any $y \in \mathbb{R}_+$, there is a unique $c \in \mathbb{R}_+$ such that g(c) = y; we denote this unique value c by h(y), so that we have h mapping from \mathbb{R}_+ to \mathbb{R}_+ and:

$$g(h(y)) = y \text{ for all } y \ge 0 \tag{27}$$

For any $c \in \mathbb{R}_+$, $g(c) \in \mathbb{R}_+$ by (15). Let us denote g(c) by x, and noting that g(c) is in the domain of h, let us denote h(g(c)) by z. Then, we get $z \in \mathbb{R}_+$, and so we evaluate g at this z, and get g(h(g(c))) = g(z). Since $g(c) \in \mathbb{R}_+$, we can apply (27) to also obtain g(h(g(c))) = g(c). Thus, we must have g(c) = g(z). Since g is one-to-one, this implies that z = c. That is, by definition of z,

$$h(g(c)) = c \text{ for all } c \ge 0 \tag{28}$$

From (28), we also see that *h* is a map from \mathbb{R}_+ onto \mathbb{R}_+ . And, from (27), we infer that *h* is strictly increasing on \mathbb{R}_+ , since *g* is strictly increasing on \mathbb{R}_+ . Next, note that *h* is continuous on \mathbb{R}_+ . To see this, let $y_1, y_2 \in \mathbb{R}_+$, with $y_1 > y_2$. Then, we have $c_1 \equiv h(y_1) > h(y_2) \equiv c_2$. By using (25) and (27), we have:

$$y_1 - y_2 = g(c_1) - g(c_2) \ge (c_1 - c_2)/(1 - \rho)$$

Thus, $0 < h(y_1) - h(y_2) \le (1 - \rho)(y_1 - y_2)$, so that *h* is Lipschitz, and hence continuous. Finally note that by (25), $g(c) \ge c$ for all $c \ge 0$, and since *h* is increasing on \mathbb{R}_+ , $h(g(c)) \ge h(c)$ for all $c \ge 0$. Thus, using (28), we have $h(c) \le c$ for all $c \ge 0$. Using Step 1, we can now infer that *h* (the inverse of *g* defined by (25)) solves the Ramsey–Euler functional equation (15).

Step 4: [The Numerical Illustration in Example 1]

In the numerical illustration in Example 1 in the text, we have chosen $a = (1/2) = \rho$. The parametric restriction on θ in (26) implies that we must have $\theta = (1/2)$. Consequently, g defined in (25) takes the form: $g(c) = 2c + c^{\frac{1}{2}}$ for all $c \ge 0$. Inverting this function (see Sect. 7.1) yields $h(y) = \frac{(1+4y)-(1+8y)^{\frac{1}{2}}}{8}$, the formula appearing in (5) in the main text.

7.3 A Useful lemma

Lemma 1 Consider a Ramsey–Euler consumption function c(y) such that

R.1 x(y) = y - c(y) is non-decreasing in y on \mathbb{R}_+ *R.2 For any interval* $[y', y''] \subset \mathbb{R}_{++}$, $\inf\{c(z) : z \in [y', y'']\} > 0$.

Further, assume that the consumption and investment processes generated by c(y) satisfy the Transversality Condition (TC). Then, c(y) is optimal.

Proof Let $Y = \mathbb{R}_+$. Fix initial stock $\tilde{y} \in Y$ with $\tilde{y} > 0$. Consider the stochastic process of output, consumption and investment $\{y_t(\tilde{y}, \omega), c_t(\tilde{y}, \omega), x_t(\tilde{y}, \omega)\}_{t=0}^{\infty}$ for $\omega \in \Omega$, hereafter written as $\{\mathbf{y}_t, \mathbf{c}_t, \mathbf{x}_t\}$, generated by the consumption function c(y). It is easy to check that $\mathbf{y}_t > 0$, $\mathbf{c}_t > 0$, $\mathbf{x}_t > 0$ for all $t \ge 0$. Equality or inequalities involving these random variables should be interpreted as holding for all $\omega \in \Omega$. Note that $\{\mathbf{y}_t, \mathbf{c}_t, \mathbf{x}_t\}$ is *feasible* from \tilde{y} . We have to establish that it is *optimal* from \tilde{y} .

Let $\{y_t\}, \{\overline{y}_t\}$ be the deterministic sequences defined by:

$$\underline{y}_0 = \overline{y}_0 = \widetilde{y}, \underline{y}_{t+1} = \underline{f}(x(\underline{y}_t)), \overline{y}_{t+1} = \overline{f}(x(\overline{y}_t)), t \ge 0.$$
(29)

Note that f(.), $\overline{f}(.)$ are nondecreasing on Y. Further, from R.1, it is easy to check that for all $\overline{t} \ge 0$:

$$\overline{y}_t \ge \mathbf{y}_t \ge \underline{y}_t \tag{30}$$

As x(z) > 0 for all z > 0 and f(x) > 0, $\overline{f}(x) < \infty$ for all x > 0,

$$0 < y_t \le \overline{y}_t < \infty \text{ for all } t \ge 0.$$
(31)

Let $\{\underline{c}_t\}, \{\overline{c}_t\}$ be the sequences defined by:

$$\underline{c}_t = \inf\{c(z) : z \in [\underline{y}_t, \overline{y}_t]\}, \ \overline{c}_t = \sup\{c(z) : z \in [\underline{y}_t, \overline{y}_t]\} \text{ for all } t \ge 0.$$
(32)

Using R.2, we have, $\underline{c}_t > 0$ for all $t \ge 0$; further, $\overline{c}_t \le \overline{y}_t < \infty$ for all $t \ge 0$. Using (30) and (32), we have:

$$\infty > \overline{c}_t \ge \mathbf{c}_t = c(\mathbf{y}_t) \ge \underline{c}_t > 0 \text{ for all } t \ge 0.$$
(33)

Thus, for every $t \ge 0$:

$$-\infty < u(\underline{c}_t) \le u(c_t) \le u(\overline{c}_t) < \infty \tag{34}$$

so that for each *t*, $u(\mathbf{c}_t)$ is a bounded \mathcal{F}_t – measurable function and has finite expectation.

Using (33), we can define the stochastic price process $\{p_t(\overline{y}, \omega)\}$, hereafter written as $\{\mathbf{p}_t\}$, by:

$$\mathbf{p}_t = \rho^t u'(\mathbf{c}_t) \text{ for } t \ge 0.$$
(35)

As before, equality or inequalities involving these random variables should be interpreted as holding for all $\omega \in \Omega$. It follows (from (33)) that for every $t \ge 0$,

$$0 < \rho^t u'(\overline{c}_t) \le \mathbf{p}_t \le \rho^t u'(\underline{c}_t) < \infty$$

i.e., \mathbf{p}_t is a bounded \mathcal{F}_t -measurable random variable (and hence integrable) for each *t*.

For all $c \ge 0$, and all $t \ge 0$, we have by concavity of u and (35),

$$\rho^{t} u(\mathbf{c}_{t}) - \mathbf{p}_{t} \mathbf{c}_{t} \ge \rho^{t} u(c) - \mathbf{p}_{t} c$$
(36)

so that for each $t \ge 0$, we have:

$$E\rho^{t}u(\mathbf{c}_{t}) - E\mathbf{p}_{t}\mathbf{c}_{t} \ge E\rho^{t}u(\widehat{\mathbf{c}}_{t}) - E\mathbf{p}_{t}\widehat{\mathbf{c}}_{t}$$
(37)

for every bounded \mathcal{F}_t measurable random variable $\hat{\mathbf{c}}_t \ge 0$ defined on Ω . Note that (using (34)), $E \rho^t u(\mathbf{c}_t)$ is finite; further, as $\hat{\mathbf{c}}_t$ is a bounded random variable, $E \rho^t u(\hat{\mathbf{c}}_t)$ on the right hand side of (37) is well defined though it may be $-\infty$.

Using the Ramsey–Euler condition (RE) and (35), one can see that ¹⁸:

$$\mathbf{p}_t = \rho^t u'(\mathbf{c}_t) = E\{\mathbf{p}_{t+1} f'(\mathbf{x}_t, r_{t+1}) | \mathcal{F}_t\}$$
(38)

Using the concavity of f (in x) we have for all $x \ge 0$ and all $t \ge 0$,

$$f(x, r_{t+1}) - f(\mathbf{x}_t, r_{t+1}) \le f'(\mathbf{x}_t, r_{t+1})(x - \mathbf{x}_t)$$

so that:

$$\mathbf{p}_{t+1}f(x, r_{t+1}) - \mathbf{p}_{t+1}f(\mathbf{x}_t, r_{t+1}) \le \mathbf{p}_{t+1}f'(\mathbf{x}_t, r_{t+1})(x - \mathbf{x}_t)$$
(39)

Thus, for every bounded \mathcal{F}_t measurable random variable $\widehat{\mathbf{x}}_t \ge 0$ defined on Ω , taking the conditional expectation with respect to \mathcal{F}_t in (39) with $x = \widehat{\mathbf{x}}_t$ we get:

$$E\{\mathbf{p}_{t+1} f(\widehat{\mathbf{x}}_t, r_{t+1}) | \mathcal{F}_t\} - E\{\mathbf{p}_{t+1} f(\mathbf{x}_t, r_{t+1}) | \mathcal{F}_t\}$$

$$\leq E\{\mathbf{p}_{t+1} f'(\mathbf{x}_t, r_{t+1}) (\widehat{\mathbf{x}}_t - \mathbf{x}_t) | \mathcal{F}_t\}$$

$$= (\widehat{\mathbf{x}}_t - \mathbf{x}_t) E\{\mathbf{p}_{t+1} f'(\mathbf{x}_t, r_{t+1}) | \mathcal{F}_t\} = \mathbf{p}_t (\widehat{\mathbf{x}}_t - \mathbf{x}_t)$$
(40)

where the third line uses the fact that $\hat{\mathbf{x}}_t$ and \mathbf{x}_t are \mathcal{F}_t measurable and the last line in (40) uses (38). Transposing terms in (40), for every bounded \mathcal{F}_t measurable $\hat{\mathbf{x}}_t \ge 0$, we have:

$$E\{\mathbf{p}_{t+1}f(\mathbf{x}_t, r_{t+1})|\mathcal{F}_t\} - \mathbf{p}_t\mathbf{x}_t \ge E\{\mathbf{p}_{t+1}f(\widehat{\mathbf{x}}_t, r_{t+1})|\mathcal{F}_t\} - \mathbf{p}_t\widehat{\mathbf{x}}_t$$
(41)

¹⁸ Strictly speaking, this involves switching from conditional expectation with respect to the distribution function F to a conditional expectation with respect to a sub sigma field.

so that:

$$E\{\mathbf{p}_{t+1}f(\mathbf{x}_t, r_{t+1})\} - E\{\mathbf{p}_t\mathbf{x}_t\} \ge E\{\mathbf{p}_{t+1}f(\widehat{\mathbf{x}}_t, r_{t+1})\} - E\{\mathbf{p}_t\widehat{\mathbf{x}}_t\}$$
(42)

Next, one can show that for *any feasible* stochastic process of output, consumption and investment $\{\widehat{\mathbf{y}}_t, \widehat{\mathbf{c}}_t, \widehat{\mathbf{x}}_t\}$ from initial stock \widetilde{y} , and for every $T \in \mathbb{N}$

$$E\left\{\sum_{t=0}^{T}\rho^{t}u(\widehat{\mathbf{c}}_{t})\right\} - E\left\{\sum_{t=0}^{T}\rho^{t}u(\mathbf{c}_{t})\right\} \le E\{\mathbf{p}_{T}\mathbf{x}_{T}\} - E\{\mathbf{p}_{T}\widehat{\mathbf{x}}_{T}\}$$
(43)

To see (43), note that from (37) we have for $t \ge 1$

$$E\rho^{t}u(\widehat{\mathbf{c}}_{t}) - E\rho^{t}u(\mathbf{c}_{t})$$

$$\leq E\mathbf{p}_{t}\widehat{\mathbf{c}}_{t} - E\mathbf{p}_{t}\mathbf{c}_{t} = [E\mathbf{p}_{t}\widehat{\mathbf{y}}_{t} - E\mathbf{p}_{t}\widehat{\mathbf{x}}_{t}] - [E\mathbf{p}_{t}\mathbf{y}_{t} - E\mathbf{p}_{t}\mathbf{x}_{t}]$$

$$= [E\mathbf{p}_{t}\widehat{\mathbf{y}}_{t} - E\mathbf{p}_{t-1}\widehat{\mathbf{x}}_{t-1}] + [E\mathbf{p}_{t-1}\widehat{\mathbf{x}}_{t-1} - E\mathbf{p}_{t}\widehat{\mathbf{x}}_{t}]$$

$$- [E\mathbf{p}_{t}\mathbf{y}_{t} - E\mathbf{p}_{t-1}\mathbf{x}_{t-1}] - [E\mathbf{p}_{t-1}\mathbf{x}_{t-1} - E\mathbf{p}_{t}\mathbf{x}_{t}]$$

$$\leq [E\mathbf{p}_{t-1}\widehat{\mathbf{x}}_{t-1} - E\mathbf{p}_{t}\widehat{\mathbf{x}}_{t}] - [E\mathbf{p}_{t-1}\mathbf{x}_{t-1} - E\mathbf{p}_{t}\mathbf{x}_{t}]$$

where the first inequality uses (37) and the second inequality uses (42).

The transversality condition (TC) implies that

$$E\{\mathbf{p}_t \mathbf{x}_t\} \to 0 \text{ as } t \to \infty \tag{44}$$

For any feasible stochastic process of output, consumption and investment $\{\widehat{\mathbf{y}}_t, \widehat{\mathbf{c}}_t, \widehat{\mathbf{x}}_t\}$ from initial stock \overline{y} ,

$$E\left\{\sum_{t=0}^{\infty}\rho^{t}u(\widehat{\mathbf{c}}_{t})\right\} - E\left\{\sum_{t=0}^{\infty}\rho^{t}u(\mathbf{c}_{t})\right\} = \lim_{T\to\infty}E\left\{\sum_{t=0}^{T}\rho^{t}u(\widehat{\mathbf{c}}_{t}) - \sum_{t=0}^{T}\rho^{t}u(\mathbf{c}_{t})\right\}$$
$$\leq \lim_{T\to\infty}\sup\left[E\{\mathbf{p}_{T}\mathbf{x}_{T}\} - E\{\mathbf{p}_{T}\widehat{\mathbf{x}}_{T}\}\right] \leq 0.$$

where the equality follows from (1), the first inequality uses (43) and the second inequality uses the transversality condition (44). Hence, c(y) is optimal. This completes the proof of the lemma.

7.4 Proof of Proposition 2

Proof We claim that under the hypothesis of Proposition 2, we always have:

R.1 x(y) is non-decreasing on \mathbb{R}_+ R.2 For any interval $[y', y''] \subset \mathbb{R}_{++}, \underline{c}(y', y'') > 0.$

R.1 and R.2 are obvious if c(y) is co-monotone. On the other hand, if c(y) is continuous, R.2 follows immediately and one can show that R.1 holds i.e., x(y) is

non-decreasing in y.¹⁹ The proof will use the properties R.1 and R.2 of the policy function.

Fix any $y_0 > 0$ and let $\{c_t\}$, $\{x_t\}$ and $\{y_t\}$ be the stochastic processes of consumption, investment and output generated by c(y) given y_0 . We will show that under the hypothesis of the proposition, $\rho^t E[u'(c_t)y_t] \rightarrow 0$ as $t \rightarrow \infty$. As $x_t \le y_t$ this implies that the transversality condition (TC) holds. As R.1 and R.2 hold, Lemma 1 then implies that $\{c_t\}$, $\{x_t\}$ and $\{y_t\}$ are optimal from y_0 ; thus c(y) is an optimal consumption function.

Recall z_1 , z_2 as defined in assumption (**T.3**)(i). There are (only) two possibilities regarding the behavior of x(y) near zero:

(A.i) There exists a sequence $\{y^n\}_{n=1}^{\infty} \to 0, y^n > 0$ for all *n* and

$$\frac{\underline{f}(x(y^n))}{y^n} \ge 1 \text{ for all } n$$

(A.ii) There exists $\hat{\epsilon} \in (0, z_1)$ such that

$$\frac{f(x(y))}{y} = \frac{f(x(y), a)}{y} < 1 \text{ for all } y \in (0, \widehat{\epsilon})$$

There are (only) two possibilities regarding the behavior of the policy function for large *y*:

(B.i) There exists a sequence $\{w^n\}_{n=1}^{\infty}$, $w^n > 0$ for all n, $\{w^n\}_{n=1}^{\infty} \to \infty$ and $\frac{\overline{f}(x(w^n))}{w^n} \le 1$ for all n.

(B.ii) There exists $\widehat{y} > z_2$ such that $\frac{\overline{f}(x(y))}{y} = \frac{f(x(y),b)}{y} > 1$ for all $y \ge \widehat{y}$.

The rest of the proof considers four cases based on combinations of these possibilities.

CASE 1: (A.i) and (B.i) hold.

There exists *N* such that $y^N \leq y_0 \leq w^N$. Fix *N*. Using R.1 and $\frac{f(x(y^N))}{y^N} \geq 1$, $\frac{\overline{f}(x(w^N))}{w^N} \leq 1$, it is easy to check that $y^N \leq y_t \leq w^N$ for all *t*. Further, using R.2, we have $c_t \geq \underline{c}(y^N, w^N) > 0$ with probability one and for all *t*. Thus,

$$0 \le E[u'(c_t)y_t] \le E[u'(\underline{c}(y^N, w^N))w^N] \text{ for all } t,$$

so that $\rho^t E[u'(c_t)y_t] \to 0$ as $t \to \infty$.

CASE 2. Suppose that the candidate policy function satisfies (A.ii) and (B.ii). As (A.ii) holds, for all $y \in (0, \hat{\epsilon})$,

$$\frac{f(x(y))}{y} = \frac{f(x(y), a)}{y} < 1$$
(45)

¹⁹ To see this, suppose $x(y^1) > x(y^2)$ for $0 \le y^1 < y^2$. Then, $x(y^1) > 0$ so that $y^1 > 0$. As x(y) = y - c(y) is continuous and x(0) = 0, there exists $y^3 \in (0, y^1)$ such that $x(y^3) = x(y^2)$. Then, $c(y^3) < c(y^2)$. $x(y^3) = x(y^2)$ implies that the right hand side of the Ramsey–Euler condition (RE) evaluated $y = y^2$ and $y = y^3$ are equal, implying $u'(c(y^3)) = u'(c(y^2))$ that contradicts $c(y^3) < c(y^2)$.

First, consider the case where $f'(0, a) > \tau_0$. Note that f'(0, a) may equal $+\infty$. Choose $\lambda_0 \in (0, 1)$ such that

$$\frac{\tau_0}{\lambda_0} < f'(0,a) \tag{46}$$

Using (45) and (46),

$$\lim_{y \to 0} \sup \frac{f(x(y), a)}{y} \frac{\tau_0}{f'(x(y), a)} \le \lim_{y \to 0} \sup \frac{\tau_0}{f'(x(y), a)} < \lambda_0$$
(47)

Next, consider the case where $f'(0, a) \le \tau_0$. Using assumption (**T.3**)(ii), $\tau_0 < \infty$ so that $f'(0, a) < \infty$. The latter implies

$$\lim_{x \to 0} \frac{f(x,a)}{f'(x,a)x} = 1$$

so that (as $\lim_{y\to 0} x(y) = 0$)

$$\frac{f(x(y), a)}{f'(x(y), a)x(y)} \to 1 \text{ as } y \to 0$$
(48)

Let $\underline{\alpha} = \liminf_{y \to 0} \frac{c(y)}{y}$. Using Condition (GP1),

$$\lim_{y \to 0} \sup \frac{x(y)}{y} = 1 - \left(\liminf_{y \to 0} \frac{c(y)}{y} \right) = 1 - \underline{\alpha} < \frac{1}{\tau_0}$$
(49)

so that

$$\lim_{y \to 0} \sup \left[\frac{f(x(y), a)}{y} \frac{\tau_0}{f'(x(y), a)} \right]$$

=
$$\lim_{y \to 0} \sup \frac{x(y)}{y} \tau_0 \left(\frac{f(x(y), a)}{f'(x(y), a)x(y)} \right)$$

=
$$(1 - \underline{\alpha})\tau_0 < 1$$
 (50)

where we use (48) and (49) in the last line. Choose λ such that

$$\lambda \in (\lambda_0, 1), \text{ if } f'(0, a) > \tau_0$$

$$\in ((1 - \underline{\alpha})\tau_0, 1) \text{ if } f'(0, a) \le \tau_0.$$

Using (47) and (50), then there exists σ and $\epsilon_1, 0 < \epsilon_1 < \hat{\epsilon}, \sigma > 0$, such that

$$\frac{f(x(y),a)}{y}\frac{(1+\sigma)\tau_0}{f'(x(y),a)} < \lambda \text{ for all } y \in (0,\epsilon_1).$$
(51)

Fix such σ , ϵ_1 . From the definition of τ_0 , there exists ϵ , $0 < \epsilon < \epsilon_1$, such that

$$\frac{\overline{\eta}(x)}{\eta(x,a)} \le (1+\sigma)\tau_0 \text{ for all } x \in (0,\epsilon)$$

so that

$$\frac{\eta(x,r)}{\eta(x,a)} \le (1+\sigma)\tau_0 \text{ for all } x \in (0,\epsilon), r \in A.$$
(52)

As (B.ii) holds, it must be the case that

$$\lim_{x \to \infty} \frac{f(x, b)}{x} \ge 1$$

so that

$$\lim_{x \to \infty} \frac{f'(x, b)x}{f(x, b)} = 1.$$
 (53)

Under assumption (**T.3**)(ii), $\tau_{\infty} < \infty$ and using condition (GP2)

$$\overline{\alpha} = \lim_{y \to \infty} \inf \frac{c(y)}{y} > 1 - \frac{1}{\tau_{\infty}}$$

so that

$$\tau_{\infty}(1-\overline{\alpha}) < 1$$

Choose $\zeta > 0, 0 < \beta < 1$ such that

$$\widetilde{\lambda} = \tau_{\infty} (1+\zeta) \frac{1-\beta \overline{\alpha}}{\beta} < 1.$$
(54)

Fix ζ , β . Using (53), there exists $\overline{y} > z_2$ such that for all $y \ge \overline{y}$

$$\frac{f'(x(y), b)x(y)}{f(x(y), b)} \ge \beta$$
$$\frac{c(y)}{y} \ge \beta \overline{\alpha},$$

and

$$\frac{\overline{\eta}(x(y))}{\eta(x(y),b)} \le (1+\zeta)\tau_{\infty}.$$

which implies that for all $y \ge \overline{y}$

$$\frac{\overline{\eta}(x(y))}{\eta(x(y),b)} \left\{ \frac{f(x(y),b)}{f'(x(y),b)x(y)} \right\} \frac{x(y)}{y} \le (1+\zeta)\tau_{\infty}(\frac{1-\beta\overline{\alpha}}{\beta}) = \widetilde{\lambda}.$$
 (55)

where $\widetilde{\lambda}$ is defined in (54).

Let E_t denote the expectation conditional on information available in period t. Observe that

$$E_{t}[u'(c(y_{t+1}))y_{t+1}] = E_{t}[u'(c(y_{t+1}))y_{t+1}I_{\{y_{t}<\epsilon\}}] + E_{t}[u'(c(y_{t+1}))y_{t+1}I_{\{y_{t}<\overline{y}\}}] + E_{t}[u'(c(y_{t+1}))y_{t+1}I_{\{y_{t}\in[\epsilon,\overline{y}]\}}]$$

Observe that

$$E_{t}[u'(c(y_{t+1}))y_{t+1}I_{\{y_{t}<\epsilon\}}]$$

$$= E_{t}\left[u'(c(y_{t+1}))f'(x_{t}, r_{t+1})y_{t}\{\frac{f(x_{t}, r_{t+1})}{y_{t}}\frac{1}{f'(x_{t}, r_{t+1})}\}I_{\{y_{t}<\epsilon\}}\right]$$

$$= E_{t}\left[u'(c(y_{t+1}))f'(x_{t}, r_{t+1})y_{t}\{\frac{x_{t}}{y_{t}}\frac{\eta(x_{t}, r_{t+1})}{\eta(x_{t}, a)}\eta(x_{t}, a)\}I_{\{y_{t}<\epsilon\}}\right]$$

$$\leq E_{t}\left[u'(c(y_{t+1}))f'(x_{t}, r_{t+1})y_{t}\{\frac{x_{t}}{y_{t}}(1+\sigma)\tau_{0}\frac{1}{\eta(x_{t}, a)}\}I_{\{y_{t}<\epsilon\}}\right], \text{ using (52)}$$

$$= E_{t}\left[u'(c(y_{t+1}))f'(x_{t}, r_{t+1})y_{t}\{\frac{f(x_{t}, a)}{y_{t}}\frac{1}{f'(x_{t}, a)}(1+\sigma)\tau_{0}\}I_{\{y_{t}<\epsilon\}}\right]$$

$$\leq \lambda E_{t}\left[u'(c(y_{t+1}))f'(x_{t}, r_{t+1})y_{t}I_{\{y_{t}<\epsilon\}}\right], \text{ using (51).}$$

Also,

$$E_{t}[u'(c(y_{t+1}))y_{t+1}I_{\{y_{t}>\overline{y}\}}]$$

$$= E_{t}\left[u'(c(y_{t+1}))f'(x_{t}, r_{t+1})y_{t}\left\{\frac{f(x_{t}, r_{t+1})}{f'(x_{t}, r_{t+1})x_{t}}\right\}\frac{x_{t}}{y_{t}}I_{\{y_{t}>\overline{y}\}}\right]$$

$$= E_{t}\left[u'(c(y_{t+1}))f'(x_{t}, r_{t+1})y_{t}\left\{\frac{\eta(x_{t}, r_{t+1})}{\eta(x_{t}, b)}\right\}\eta(x_{t}, b)\frac{x_{t}}{y_{t}}I_{\{y_{t}>\overline{y}\}}\right]$$

$$\leq E_{t}\left[u'(c(y_{t+1}))f'(x_{t}, r_{t+1})y_{t}\frac{\overline{\eta}(x_{t})}{\eta(x_{t}, b)}\left\{\frac{f(x(y_{t}), b)}{f'(x(y_{t}), b)x(y_{t})}\right\}\left\{\frac{x(y_{t})}{y_{t}}\right\}I_{\{y_{t}>\overline{y}\}}\right]$$

$$\leq \widetilde{\lambda}E_{t}\left[u'(c(y_{t+1}))f'(x_{t}, r_{t+1})y_{t}I_{\{y_{t}>\overline{y}\}}\right], \text{ using (55)}$$
(57)

Finally, given fixed $\epsilon, \overline{y} \in \mathbb{R}_{++}$ as defined above, $y_t \in [\epsilon, \overline{y}]$ implies that with probability one, $y_{t+1} \in [\underline{f}(x(\epsilon)), \overline{f}(\overline{y})] \subset \mathbb{R}_{++}$. Therefore for all *t*,

$$E_{t}[u'(c(y_{t+1}))y_{t+1}I_{\{y_{t}\in[\epsilon,\overline{y}]\}}]$$

$$\leq u'(\underline{c}(f(x(\epsilon)),\overline{f(\overline{y})}))\overline{f(\overline{y})} = Q'.$$
(58)

where $0 < Q' < \infty$, using R.2. Let $\hat{\lambda} = \max\{\lambda, \tilde{\lambda}\}$. Then, $\hat{\lambda} \in (0, 1)$ and using (56),(57) and (58)

$$\rho^{t+1}E_{t}[u'(c(y_{t+1}))y_{t+1}] = \rho^{t+1}E_{t}[u'(c(y_{t+1}))y_{t+1}I_{\{y_{t}<\epsilon\}}] + \rho^{t+1}E_{t}[u'(c(y_{t+1}))y_{t+1}I_{\{y_{t}>\overline{y}\}}] + \rho^{t+1}E_{t}[u'(c(y_{t+1}))y_{t+1}I_{\{y_{t}\in[\epsilon,\overline{y}]\}}]$$

$$\leq \rho^{t+1}\widehat{\lambda}[E_{t}[u'(c(y_{t+1}))f'(x_{t},r_{t+1})y_{t}(I_{\{y_{t}<\epsilon\}}+I_{\{y_{t}>\overline{y}\}})] + \rho^{t+1}Q'$$

$$\leq \widehat{\lambda}\rho^{t+1}E_{t}[u'(c_{t+1})f'(x_{t},r_{t+1})]y_{t} + \rho^{t+1}Q'$$

$$= \widehat{\lambda}\rho^{t}u'(c_{t})y_{t} + \rho^{t+1}Q'$$
(59)

where the last equality follows from the Ramsey–Euler equation (RE). Taking unconditional expectation in (59) we have:

$$\rho^{t+1} E[u'(c(y_{t+1}))y_{t+1}] \\ \leq \hat{\lambda} \rho^t E[u'(c_t)y_t] + \rho^{t+1} Q'$$

for all *t*, which implies that $\rho^t E[u'(c_t)y_t] \to 0$ as $t \to \infty$.

CASE 3. Suppose that the candidate policy function satisfies (A.i) and (B.ii).

As in CASE 1, as (A.i) holds, there exists N such that $y^N \le y_t$ with probability one for all t. As (B.ii) holds, using an identical argument as in CASE 2 (and with some abuse of notation), condition (GP2) ensures that there exists $0 < \beta < 1, \zeta > 0, \overline{y} > 0$ such that

$$0 < \widetilde{\lambda} = (1 + \zeta)\tau_{\infty}(\frac{1 - \beta\overline{\alpha}}{\beta}) < 1$$

and

$$\frac{\overline{\eta}(x(y))}{\eta(x(y),b)} \left\{ \frac{f(x(y),b)}{f'(x(y),b)x(y)} \right\} \frac{x(y)}{y} \le \widehat{\lambda} \text{ for all } y > \overline{y}$$

so that

$$E_{t}[u'(c(y_{t+1}))y_{t+1}I_{\{y_{t}>\overline{y}\}}] \\ \leq \tilde{\lambda}E_{t}\left[u'(c(y_{t+1}))f'(x_{t}, r_{t+1})y_{t}I_{\{y_{t}>\overline{y}\}}\right]$$

Further,

$$E_t[u'(c(y_{t+1}))y_{t+1}I_{\{y_t\in[\epsilon,\overline{y}]\}}] \le u'(\underline{c}(f(x(y^N)),\overline{f}(\overline{y}))))\overline{f}(\overline{y}) = Q''.$$

where $0 < Q'' < \infty$. Thus

$$E_{t}[u'(c(y_{t+1}))y_{t+1}] = E_{t}[u'(c(y_{t+1}))y_{t+1}I_{\{y_{t} > \overline{y}\}}] + E_{t}[u'(c(y_{t+1}))y_{t+1}I_{\{y_{t} \in [y^{n}, \overline{y}]\}}]$$

$$\leq \tilde{\lambda}E_{t}[u'(c_{t+1})f'(x_{t}, r_{t+1})]y_{t} + Q''$$

and the rest of the proof is identical to CASE 2.

CASE 4. Suppose that the candidate policy function satisfies (A.ii) and (B.i).

As (B.i) holds, using an identical argument as in the proof for CASE 1, there exists *n* such that $y_t \le w^n$ with probability one for all *t*. As condition (A.ii) holds, arguments identical to those used in Case 2 imply that (with some abuse of notation) there exists $\epsilon, \lambda, 0 < \epsilon < \hat{\epsilon}, 0 < \lambda < 1$ such that

$$\frac{\eta(x(y), r)}{\eta(x(y), a)} \le (1 + \sigma)\tau_0,$$

$$\frac{f(x(y), a)}{y} \frac{(1 + \sigma)\tau_0}{f'(x(y), a)} < \lambda \text{ for all } y \in (0, \epsilon).$$

and therefore, using identical arguments leading to (56), we have

$$E_t[u'(c(y_{t+1}))y_{t+1}I_{\{y_t<\epsilon\}}] \le \lambda E_t\left[u'(c(y_{t+1}))f'(x_t, r_{t+1})y_tI_{\{y_t<\epsilon\}}\right].$$

Further,

$$E_t[u'(c(y_{t+1}))y_{t+1}I_{\{y_t \ge \epsilon\}}] = E_t[u'(c(y_{t+1}))y_{t+1}I_{\{\epsilon \le y_t \le w^n\}}]$$

$$\leq u'(c(f(x(\epsilon)), \overline{f}(x(w^n)))\overline{f}(w^n) = Q^{\wedge}.$$

where $0 < Q^{2} < \infty$ using R.1. Then,

$$E_{t}[u'(c(y_{t+1}))y_{t+1}] = E_{t}[u'(c(y_{t+1}))y_{t+1}I_{\{y_{t}<\epsilon\}}] + E_{t}[u'(c(y_{t+1}))y_{t+1}I_{\{\epsilon\leq y_{t}\leq w^{n}\}}] \leq \lambda E_{t}[u'(c_{t+1})f'(x_{t}, r_{t+1})]y_{t} + Q^{2},$$

and the rest of the proof is identical to Case 2. This completes the proof of Proposition 2. $\hfill \Box$

7.5 Proof of Proposition 3

Proof Using identical arguments as at the beginning of the proof of Proposition 2, one can show that properties R.1 and R.2 in the antecedent of Lemma 1 hold. Fix any $y_0 > 0$ and let $\{c_t\}, \{x_t\}$ and $\{y_t\}$ be the stochastic paths of consumption, investment and output generated by c(y) given y_0 . We will show that under the hypothesis of the proposition, $\rho^t E[u'(c_t)x_t] \rightarrow 0$ as $t \rightarrow \infty$. Lemma 1 then implies that $\{c_t\}, \{x_t\}$ and $\{y_t\}$ are optimal from y_0 ; thus c(y) is an optimal consumption function. Let $\hat{\alpha} > 0$ be defined by

$$\inf_{y>0}\frac{c(y)}{y} = \hat{\alpha}$$

Then, $\widehat{\alpha} \in (0, 1)$. Observe that for any *t*:

$$\rho^{t} E[u'(c_{t})x_{t}] \leq \rho^{t} E[u'(\alpha y_{t})y_{t}] \leq \rho^{t} E[u'(\alpha y_{t})y_{t}] = \rho^{t} \left[\frac{E(u'(\alpha y_{t})\alpha y_{t})}{\widehat{\alpha}} \right] \leq \rho^{t} \left[\frac{M}{\widehat{\alpha}} \right]$$

which converges to 0 as $t \to \infty$.

References

Acemoglu, D.: Introduction to Modern Economic Growth. Princeton University Press, Princeton (2009) Benhabib, J., Rustichini, A.: A note on a new class of solutions to dynamic programming problems arising

in economic growth. J. Econ. Dyn. Control 18, 807-813 (1994)

Brock, W.A., Majumdar, M.: On characterizing optimal competitive programs in terms of decentralizable conditions. J. Econ. Theory 45, 262–73 (1988)

- Brock, W.A., Mirman, L.J.: Optimal economic growth under uncertainty: the discounted case. J. Econ. Theory **4**, 479–513 (1972)
- Cass, D.: Optimum growth in an aggregative model of capital accumulation. Rev. Econ. Stud. **32**, 233–40 (1965)
- Dasgupta, S., Mitra, T.: Characterization of intertemporal optimality in terms of decentralizable conditions. J. Econ. Theory 45, 274–87 (1988)
- de Hek, P.: On endogenous growth under uncertainty. Int. Econ. Rev. 40, 727–44 (1999)
- de Hek, P., Roy, S.: On sustained growth under uncertainty. Int. Econ. Rev. 42, 801-14 (2001)
- Gale, D., Sutherland, W.R.: Analysis of a one-good model of economic development. In: Dantzig, G., Veinott, A. (eds.) Mathematics of the Decision Sciences, Part 2, American Mathematical Society in Applied Mathematics, vol. 12. American Mathematical Society, Providence (1968)
- Jones, L.E., Manuelli, R.E.: A convex model of equilibrium growth: theory and policy implications. J. Polit. Econ. 98, 1008–38 (1990)
- Jones, L.E., Manuelli, R.E., Siu, H.E., Stacchetti, E.: Fluctuations in convex models of endogenous growth I: growth effects. Rev. Econ. Dyn. 8(4), 780–804 (2005)
- Kamihigashi, T.: Necessity of transversality conditions for infinite horizon problems. Econometrica 69, 995–2012 (2001)
- Kamihigashi, T.: Necessity of transversality conditions for stochastic problems. J. Econ. Theory 109, 140– 149 (2003)
- Kamihigashi, T.: Necessity of the transversality condition for stochastic models with bounded or CRRA utility. J. Econ. Dyn. Control 29, 1313–29 (2005)
- Kamihigashi, T.: Stochastic optimal growth with bounded or unbounded utility and with bounded or unbounded shocks. J. Math. Econ. 43, 477–500 (2007)
- Kamihigashi, T.: Elementary results on solutions to the Bellman equation of dynamic programming: existence, uniqueness, and convergence. Economic Theory 56, 251–73 (2014). https://doi.org/10.1007/ s00199-013-0789-4
- Levhari, D., Srinivasan, T.N.: Optimal savings under uncertainty. Rev. Econ. Stud. 36, 153-63 (1969)
- Majumdar, M., Zilcha, I.: Optimal growth in a stochastic environment: some sensitivity and turnpike results. J. Econ. Theory 43, 116–33 (1987)
- Malinvaud, E.: Capital accumulation and efficient allocation of resources. Econometrica 21, 233–268 (1953)
- Matkowski, J., Nowak, A.S.: On discounted dynamic programming with unbounded returns. Economic Theory 46, 455–74 (2011). https://doi.org/10.1016/j.jmaa.2010.08.073
- Mirman, L.J., Zilcha, I.: On optimal growth under uncertainty. J. Econ. Theory 11, 329–339 (1975)
- Mitra, T., Roy, S.: Optimality of Ramsey-Euler policy in the stochastic growth model. J. Econ. Theory 172, 1–25 (2017)
- Mitra, T., Sorger, G.: Extinction in common property resource models: an analytically tractable example. Economic Theory 57, 41–57 (2014). https://doi.org/10.1007/s00199-013-0799-2
- Nishimura, K., Stachurski, J.: Stability of stochastic optimal growth models: a new approach. J. Econ. Theory 122, 100–118 (2005)
- Nyarko, Y.: On characterizing optimality of stochastic competitive processes. J. Econ. Theory 45, 316–329 (1988)
- Olson, L.J.: Stochastic growth with irreversible investment. J. Econ. Theory 47, 101-129 (1989)
- Peleg, B., Ryder, H.E.: On optimal consumption plans in a multi-sector economy. Rev. Econ. Stud. 39, 159–69 (1972)
- Rincón-Zapatero, J.P., Rodriguez-Palmero, C.: Existence and uniqueness of solutions to the Bellman equation. Econometrica 71, 1519–1555 (2003)
- Shell, K.: Application of Pontryagin's maximum principle to economics. In: Lecture Notes in Operations Research and Mathematical Economics, volume 11, Springer Verlag, Berlin (1969)
- Stachurski, J.: Stochastic optimal growth with unbounded shock. J. Econ. Theory 106, 40–65 (2002)
- Stokey, N., Lucas, R.E.: Recursive Methods in Economic Dynamics. Harvard University Press, Cambridge (1989)
- Weitzman, M.: Duality theory for infinite horizon convex models. Manag. Sci. 19, 783–89 (1973)
- Zilcha, I.: Characterization by prices of optimal programs under uncertainty. J. Math. Econ. **3**, 173–181 (1976)

Zilcha, I.: Transversality condition in a multi-sector economy under uncertainty. Econometrica **46**, 515–525 (1978)

Publisher's Note Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.